Reducing Occurrence of *Giardia duodenalis* in Children Living in Semiarid Regions: Impact of a Large Scale Rainwater Harvesting Initiative



Jacqueline Evangelista Fonseca¹, Mariângela Carneiro², João Luiz Pena¹, Enrico A. Colosimo³, Nívea Bispo da Silva³, André Gabriel F. C. da Costa³, Luciano E. Moreira⁴, Sandy Cairncross⁵, Léo Heller¹*

1 School of Engineering; Department of Sanitary and Environmental Engineering – Universidade Federal de Minas Gerais, Belo Horizonte, Minas Gerais, Brazil, 2 Department of Parasitology; Institute of Biological Sciences and Graduate Program in Health Sciences: Infectious Diseases and Tropical Medicine – Universidade Federal de Minas Gerais, Belo Horizonte, Minas Gerais, Brazil, 3 Department of Statistics; Institute of Mathematical Sciences – Universidade Federal de Minas Gerais, Belo Horizonte, Minas Gerais, Brazil, 4 Laboratory of Parasitology – Universidade Presidente Antônio Carlos, Teófilo Otoni, Minas Gerais, Brazil, 5 London School of Hygiene and Tropical Medicine, London, United Kingdom

Abstract

Background: In Brazil, about two million people living in rural semiarid regions were benefited with the construction of rainwater cement cisterns, as an initiative from the program "One Million Cisterns" (P1MC). Nevertheless, few epidemiological studies have been conducted to assess health risks or protection effects associated with consumption of this water source. The aim of this study was to evaluate whether access to rainwater harvesting cisterns is associated with the decrease in the occurrence of *Giardia duodenalis* infections in children, compared to other children living in households supplied by other water sources.

Methodology/Principal Findings: A quasi-experimental study with two concurrent cohorts was developed in two rural municipalities of the semiarid region of Brazil. A sample of 664 children, aged between 4 months and 5 years old, was followed up, of which 332 had access to rainwater cisterns (cistern group) and 332 did not, having water supplied from alternative sources (comparison group). In a period of approximately one year (2010) intestinal parasites were investigated in feces three times. The prevalence of *G. duodenalis* in children from the cistern group ranged from 4.8 to 10.5%, while the prevalence in the comparison group ranged from 7.6 to 16.7%. Multivariate analysis (GEE) showed a higher risk of *G. duodenalis* infection in children who did not have access to rainwater cisterns, when compared to children who did (OR 1.72; 95% CI 1.14–2.59). The other variables associated with *G. duodenalis* infection were: number of rooms per house (OR 0.89; 95% CI 0.80–0.99); family income (OR0.48; 95% CI 0.26–0.88); birth order (OR 1.72; 95% CI 1.17–2.51); preterm children (OR 1.70; 95% CI 1.19–2.43); and improper hand hygiene prior to food preparation (OR 4.78; 95% CI 1.95–11.76).

Conclusions/Significance: Ownership of a rainwater cistern is associated with a lower prevalence of *G. duodenalis* infection in children after adjustment for environmental and family-related factors. Nevertheless, the study suggests the necessity to complement physical interventions with actions related to personal and domestic hygiene to enable further reductions in parasite infections affecting mainly the underprivileged populations.

Citation: Fonseca JE, Carneiro M, Pena JL, Colosimo EA, Silva NBd, et al. (2014) Reducing Occurrence of *Giardia duodenalis* in Children Living in Semiarid Regions: Impact of a Large Scale Rainwater Harvesting Initiative. PLoS Negl Trop Dis 8(6): e2943. doi:10.1371/journal.pntd.0002943

Editor: Justin V. Remais, Emory University, United States of America

Received March 24, 2013; Accepted May 1, 2014; Published June 19, 2014

Copyright: © 2014 Fonseca et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This work was supported by the National Council for Scientific and Technological Development (CNPq Brazil). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: heller@desa.ufmg.br

Introduction

Intestinal infections caused by *Giardia duodenalis* (syn. *G. intestinalis*, *G. lamblia*) involve several different mechanisms of transmission and risk factors [1–3]. These include poor conditions of water supply and sanitation, which explain the higher prevalence of this parasite in developing countries in which access to water supply, sanitation and health services is less widespread and less efficient. Giardiasis is still considered a neglected disease by the World Health Organization, despite its significance and

scope, since it occurs primarily in the poorest regions, has a negative impact on child health, pregnancy and worker productivity, and impairs the ability of those infected to achieve their full potential, affecting their socio-economic development [4,5].

In Brazil, such intestinal parasitic infections are still a significant public health problem, despite the reversal of the mortality profile in recent decades. Infectious and parasitic diseases were the eighth most significant category of causes of death in Brazil in the year 2008, representing 4.4% of the total [6]. Enteric infections can have a long-term impact on child growth and development, and

Author Summary

Currently, rainwater harvesting cisterns built up through the "One Million Cisterns" Program, coordinated by the Brazilian Government, represents the most important public policy for water supply in the rural semiarid region of Brazil. Despite the extensive use of this water provision, few studies have been performed to assess the associated health impact. To assess such impact, this study investigated factors associated with the occurrence of Giardia duodenalis (protozoan closely related to the ingestion of contaminated water) in children aged between 4 months and five years old. Two distinct groups of children, one with access to rainwater cisterns (cistern group) and the other without cisterns, with water supplied from alternative sources, including rivers, springs or dams (comparison group) were followed up for approximately one year. It was observed that the risk occurrence of G. duodenalis prevalence was greater in children from the comparison group, when compared with those from the cistern group, indicating that the odds of infection by this parasite were 1.72 times greater for comparison group children. In conclusion, the study showed health benefits related to rainwater cisterns, but also identified that other complementary sanitation interventions are needed to minimize the risk of disease in populations with limited access to drinking water.

Giardia in particular has been shown to have an impact even when it is not causing any diarrhea at all (5).

Based on research in Brazilian communities living in poor conditions with inadequate or insufficient health and sanitation services, the prevalence of intestinal parasites ranged from 53% to 83%, while the prevalence of *G. duodenalis* ranged from 11% to 32% [7–10]. In all of these studies, the authors related the high prevalences found to the poor hygiene and water and sanitation conditions, including access to water with improper quality.

One of the forms of *G. duodenalis* transmission is the intake of water or food contaminated with protozoan cysts, which are released in the host's feces. Heller et al. [11] reported high concentrations of Giardia cysts in sanitary sewers and in water supply sources in Brazil that were contaminated by human feces and feces from animals surrounding the river basin. Such factors favor the risk of *G. duodenalis* infection, associated with water consumption. In rural areas, the situation is even more critical, considering the difficulties in accessing treated water in sufficient amounts to maintain proper hygienic conditions. In Brazil, 72% of the population living in rural areas, or approximately 21 million inhabitants, relies on alternative supply sources, such as rivers, springs and dams because they are not connected to the main water supply network [12].

In Brazilian semiarid regions, where the dry season lasts approximately eight months, the "Programme for One Million Cisterns" (P1MC) was created to enable a better quality of life for rural families, which are the ones most affected by drought. The programme was developed in 1999 by civil society organizations, which had formed a coalition known as the ASA (Articulação do Semiárido Brasileiro). Its primary goal is to build one million cement cisterns for the storage of rainwater collected through gutters attached to the roofs of homes.

In municipalities where P1MC is implemented, municipal commissions are formed by public managers, representatives of rural communities and ASA. These commissions select the families to receive the cisterns based on pre-established criteria. Families from the rural zone of the Brazilian semiarid region with no drinkable water in the surroundings of their household or with unreliable existing sources are considered top priority. The remaining criteria are evaluated in the following order [13]: (1) per capita monthly family income up to half a minimum wage (approximately USD 339), or whose total income is up to three times the minimum wage a month (approximately USD 1,017); (2) female-headed households; (3) greatest number of children less than 6 years old; (4) greatest number of children of school age; (5) greatest number of disabled people; (6) greatest number of elderly.

Each selected family receives a cistern with a capacity to store 16,000 liters of water, which, according to the ASA, is sufficient for a family composed of 5 people to drink, cook and brush their teeth during the dry season. By mid-2012, more than 418,000 cisterns had been built, which benefited approximately 2.1 million inhabitants of semiarid regions [14].

Although rainwater is recognized as a purer source of water [15], various geographic [16,17], atmospheric [18], harvesting surface-related [19–23] and cistern management-related [15,20,24,25] factors may impair its quality and, consequently, pose a risk to the health of populations that use the rainwater, especially for drinking.

Studies conducted in various parts of the world, including Australia [15,26–28], New Zealand [29], Denmark [30], the United States [31] and Brazil [32] investigated the occurrence of *G. duodenalis* cysts in rainwater harvesting tanks. These studies found between 0% and 45% positivity for *G. duodenalis* among the samples tested. Nevertheless, few epidemiological studies have been conducted to assess health risks or protective effects associated with consumption of this water source [33]. Furthermore, there are no reports in the literature of epidemiological studies of human giardiasis associated with the consumption of harvested rainwater in developing countries.

Thus, given the scope of the P1MC Programme and considering the need to investigate its effects on the health of those who benefit, this study sought to evaluate whether access to rainwater harvesting cisterns is associated with a lower prevalence of G. duodenalis infections in children under five years of age, compared to children living in households supplied by other water sources.

Methods

Ethical considerations

This study was approved by the institutional review board of the Federal University of Minas Gerais (No. ETIC 279/09). Legal guardians of the children involved in this study were required to sign an informed consent form. Furthermore, the children were granted medical care and treatment when necessary.

Study design and population

This study is part of a broader research project that sought to perform an epidemiological evaluation of the P1MC program to assess the impact of rainwater harvesting cisterns on different health indicators of families in rural communities. For this purpose, we chose to study children of up to five years of age, who are more susceptible to diseases facilitated by poor water supply and sanitation conditions. For the evaluation, in addition to the effect on the occurrence of *G. duodenalis* cysts in the children's feces, other outcomes used were the occurrence of other parasites and diarrhea [34]. Analyses of the microbiological quality of the water consumed by the monitored families were also performed [35].

The municipalities selected for conducting this study were Berilo and Chapada do Norte, which both are located in the Jequitinhonha Valley, in the semiarid northeast of Minas Gerais state, Brazil. These municipalities present a low human development index (HDI) with 0.680 for Berilo and 0.641 for Chapada do Norte. The index of Minas Gerais state is 0.773 and that of Brazil is 0.766 [36].

The study followed a quasi-experimental design with two concurrent cohorts, which are characterized as follows: <u>cistern</u> <u>group</u>: composed of children, aged 4 months to 5 years old, living in a rural area of the selected region, who had in their homes or were using third-party cisterns as rainwater storage systems; <u>comparison group</u>: composed of children, aged 4 months to 5 years old, who did not have access to a cistern. The water they used mainly originated from water sources with no sanitary protection, including rivers, springs and dams.

The sample size was established for the larger study, which included several other outcomes apart from prevalence of *G. duodenalis.* Based on an expected prevalence of the main outcome of 20% for the comparison group and 12% for the cistern group with a 95% confidence interval and 80% power of the test, an initial sample of 706 children with 353 in each group was chosen, using Epi Info software version 3.5.1. Allowing losses of 20%, this would give 565 children. Considering a prevalence of *G. duodenalis* infection of 22%, based on a range from 18 to 26% in Minas Gerais State [2,10], this sample would be sufficient to detect a reduction of prevalence to 12.8% (RR = 1.72).

In order to form the cistern group, all the children from the two municipalities which met the established requirements for the group were selected. The children from the comparison group were selected from rural communities in the same municipalities, based on the same criteria as the cistern group, with the single exception that they did not have a cistern; in practice almost all eligible children were selected. Both groups of children were supported by the same field workers of the Family Health Program.

Data collection

Field surveys were carried out during a period of approximately one year (2010), and the data collection relied on the support of community health workers from the Family Health Program, which is a component of the Brazilian Unified Health System. Health workers monitored families in a given geographic area, performing monthly visits to the houses of those families to prevent or monitor the population's health problems.

The first contact with the selected families occurred from September to December 2009. After being informed about the research aims and questioned on their interest to participate in the study, two families refused to participate. For the other families, after joining the study, a structured questionnaire was applied, including 55 common questions to all participants from both groups. From these, 48 produced categorical variables and 7 elicited continuous numerical variables, which referred to: (i) conditions of mother and child's health: child's sex and age, mother's age, child's hospitalization and vaccination, breastfeeding duration, vitamin supplementation, anti-parasite treatment, diarrhoea in the past 72 hours, pregnancy order, and prenatal exams; (ii) family structure: child's caretaker and caretaker's schooling; (iii) family socioeconomic conditions: number of rooms and inhabitants in the household, toilets, house characteristics (walls, roof and floor), if the family is given any financial aid from the government, and total family income; (iv) environmental and sanitary conditions of the household: place where the family defecates, destination of dirty diapers and wastewater, existence of a stream close to the household, and children's contact with the stream; (iv) solid waste and vectors: garbage destination, presence of flies,

cockroaches and rats in the household; (v) child and family's hygiene conditions: frequency of children's bath, habit of washing children's hands, habit of washing the food preparer's hands; disinfection of fruit and vegetables.

To assess the impact of rainwater cisterns on children's health, monitoring forms were provided in two additional stages (middle and end) after the beginning of the data collection to verify if any family from the comparison group had received a new rainwater harvesting cistern during the year in which the children were monitored. These forms were applied by health workers in April/ May 2010 and October 2010.

Parasitological evaluation

The monitoring of intestinal parasitic diseases in the children was performed in three stages at dates close to the application of the structured questionnaires and monitoring forms. The TF-Test (Three Fecal Test) method was used. It utilizes centrifugation, filtration and sedimentation and has a high sensitivity in detecting protozoa cysts [37]. This method was also selected because it allows a prolonged sample preservation time (30 days) without the need for refrigeration. In each stage, three fecal samples were collected from the same subject on alternate days in a 5 day period and placed in separate vials. This approach increased the sensitivity of the technique as appearance of Giardia in stools can be intermittent [38].

All three collection vials from each kit were identified with the child's first name to avoid confusion between fecal samples in households where more than one child participated in the research study. In the first stage, the TF-Test kits were handed out to each household during the structured questionnaires implementation. In the second and third stages, the kits were left with the health workers to be delivered to families before each specified data collection round. The health worker arranged an approximate date for sample collection with the children guardian following an orientation on the proper way of performing the sampling.

It is noteworthy that 37 children (23 in cistern group and 14 in comparison group) did not receive the TF-Test kits at Stage 1 because they were less than four months of age at the time. It was a criterion of the study not to include children up to four months of age due to the protection provided by breast milk and the very low exposure to water sources.

In Chapada do Norte, logistic field problems prevented the third stage of data collection from taking place six months after the second stage, in October 2010. The stool exams for the 416 children from the municipality, who had been monitored in stages 1 and 2, were therefore not carried out until August 2011 (16 months after the second stage). Analyses were performed by the team from the Laboratory of Parasitology, President Antônio Carlos University (Universidade Presidente Antônio Carlos; UNIPAC) in the municipality of Teófilo Ottoni, which is located approximately 270 km from Berilo and is the region's reference laboratory. The laboratory staff were blinded as to the exposure status of each child.

The results forwarded by the laboratory staff were immediately transferred to the health workers in the selected municipalities, who were responsible for providing chemotherapy to infected children.

Statistical analysis

Generalized Estimating Equations (GEE) were used for the data analysis. Household was the first cluster specification for considering more than one child in each house. Chronological stage of the study was the second level to assess the occurrence of *G. duodenalis* in each child at each time point. A logit link function was used in order to have odds ratio interpretation for the model parameters.

Exposure variables were initially considered one by one. Out of all the categorical and continuous covariates, those showing a significance level of less than 0.20 in univariate analysis were selected for the initial model. In order to reach the final model, covariates screened in the initial model (see Supporting File – Table S1) were consecutively dropped until all of them showed a significance of p < 0.05. Analyses were performed with R opensource statistical software, version 2.13.

Results

Characterization and comparison of the groups

The monitoring forms, filled out at stages 2 and 3, showed that the median household in the cistern group had owned their cistern for 36 months; no family from the comparison group had received a cistern at stage 2, and five families had received cisterns at stage 3. In the comparison group, 38 (11%) monitoring forms at stage 2 and 83 (25%) for stage 3 were neither returned nor filled out. It was assumed that the main water supply for these children at Stage 1 remained unaltered in all subsequent stages. In the analysis, out of 664 children monitored, 332 had rainwater cisterns and 332 did not at stages 1 and 2. However, due to these changes in exposure status these values became 337 and 327 children, respectively, at stage 3.

For characteristics evaluated, including those regarding child and mother's health, hygienic and sanitary habits and also concerning families' socioeconomic conditions, both groups are rather similar and there were no statistically significant differences when comparing them, besides the difference in the main water supply and in some factors associated with it, such as use of treated water or existence of river or stream close to the household. The groups are compared with regard to a selection of variables in Table 1.

Regarding the specific characteristics of the cistern group, it was found that, although 72% of the rainwater cisterns were built by

```
Table 1. Characteristics of children from cistern group compared with children from comparison group.
```

Variables		Group	Group ^a			
		Cisterr	group	Comparison group		- p-value
		N	%	N	%	_
Child's sex	Female	161	48.5	154	46.4	0.59
	Male	171	51.5	178	53.6	
Pregnancy order	First	115	34.7	116	34.9	0.96
	Second or superior	216	65.3	216	65.1	
Breastfed at least once	Yes	327	98.8	323	97.3	0.16
	No	4	1.2	9	2.7	
Prenatal monitoring during pregnancy	Yes	319	98.2	322	97.6	0.61
	No	6	1.8	8	2.4	
Duration of pregnancy	Full term	241	73.7	257	78.6	0.14
	Less than full term	86	26.3	70	21.4	
Person responsible for child's care during the last year	Mother	282	84.9	298	89.8	0.06
	Another person	50	15.1	34	10.2	
Child's caretaker's schooling	Illiterate	25	7.6	36	10.8	0.15
	Reads and writes	304	92.4	296	89.2	
Hand hygiene prior to food preparation	Always	328	98.8	324	97.6	0.24
	With small frequency or never	4	1.2	8	2.4	
House construction material	Brickwork	113	34.0	123	37.0	0.63
	Adobe (clay)	212	63.9	204	61.4	
	Wood	7	2.1	5	1.5	
Financial aid from the government	Yes	240	72.3	238	71.9	0.91
	No	92	27.7	93	28.1	
Total family income	From R\$ 0.00 to 100.00	27	8.3	25	7.7	0.94
	From R\$ 101.00 to 500.00	202	62.3	202	62.2	
	Above R\$ 500.00	95	29.3	98	30.2	
Destination of home garbage	Burnt, buried or collected	290	87.3	281	84.6	0.32
	Open-air or thrown in the river	42	12.7	51	15.4	
Flies/mosquitoes observed in the household throughout the year	Yes	296	89.2	307	92.7	0.11
	No	36	10.8	24	7.3	

^aThe children's classification was considered in the first stage (332 children in cistern group and 332 children in comparison group). doi:10.1371/journal.pntd.0002943.t001 P1MC, there were also cisterns built by other institutions (21%) or at family's expense (4%). For 3% of the children, the family could not confirm who was responsible for the construction of the cisterns. Most (85%) of the cisterns presented a total volume of 16,000 litres, 10% had a 25,000 litre capacity and the others presented different volumes (5%) or the participants could not answer (0.6%). It is important to point out that, at the time of the structured questionnaires application, there were cisterns built at least one month and at most ten years ago. On average, the cisterns were 33 months old.

For families in the comparison group who had not received cisterns, the reason was usually that they depended on other sources of water supply/75% reported that they used water from rivers, 23% used water from springs and 2% used water from dams or wells. For 79% of the children, the water is piped from the source to the inside of the house through an electric pump or by gravity, and for 15%, carried in a bucket or other container. For 6% of the children, the water is brought through a hose or there is a connection with the main water supply network. Most of the interviewed families (98%) claimed to use the water for several purposes, such as drinking, cooking, bathing, brushing teeth, washing dishes and clothes, and cleaning the house.

Epidemiological evaluation of G. duodenalis infection

In the present study, the loss to follow-up in the parasitological examinations increased at stages 1 and 2, but only slightly at stage 3. Those losses were mostly related to the family's moving to another region (migration seeking better work conditions, which is a frequent practice among families living in Brazilian and other semiarid regions) or a failure to return the TF-Test kit with the children's feces. For the cistern group and for the comparison group these losses have been, respectively: 40 and 48 in stage 1; 93 and 105 in stage 2, and 98 and 102 in stage 3.

Comparing those children that returned at least one sample for parasitological examination during the three stages, with those who did not return any, no statistical difference was found. Table 2 shows the comparison for selected variables. As shown in Table 3, the prevalence of *G. duodenalis* at Stages 1 and 2 was higher for children from households without access to cisterns, but there was a reversal at Stage 3. The prevalence in children from the cistern group ranged from 4.8% to 10.5%, while the prevalence in the comparison group ranged from 7.6% to 16.7%.

The categorical and continuous variables selected in the univariate model are presented in Tables 4 and 5 respectively. The variables that remained significant in the multivariate model after adjusting for other variables are shown in Table 6. It was observed that the odds of G. duodenalis infection were 1.72 times higher (95% CI 1.14-2.59) in children who did not have access to cisterns than among those who had access; therefore, cisterns were found to have a protective effect. This odds ratio is the same as the value which emerged from the power calculation, but this is entirely coincidental. The power calculation was based on a far higher assumed prevalence than was found for giardiasis. However, other variables were noticeably also independently associated with G. duodenalis infections. An increase in the number of rooms per house significantly reduced the odds of infection (OR 0.89; 95% CI 0.80-0.99). Regarding "birth order", firstborn children had lower odds of infection than those from subsequent births (OR 1.72; 95% CI 1.17-2.51). The odds of infection were 1.70 (95% CI 1.19-2.43) times greater for preterm children than for others. Improper hand hygiene prior to food preparation increased the odds of G. duodenalis infections by 4.78 (95% CI 1.95-11.76) times. An increased family income reduced the odds of children presenting with G. duodenalis infections (OR 0.48; 95% CI 0.26-0.88).

Discussion

The lowest prevalence of *G. duodenalis* infections in the present study, which was equivalent to 4.8%, was found in children from the cistern group at stage 1, while the highest value, corresponding to 16.7%, was found at stage 2 in children from the comparison group, that is, those children whose water supply originated from water sources without sanitary protection, including rivers, springs

Table 2. Losses regarding the return of stool samples for parasitological examination.

Variables	Category	at least	Children who returned at least one sample (n=635)		Children who returned no sample (n=29)	
		n	%	n	%	
Cistern	Yes (Cistern group)	323	50.9	14	48.3	0.79
	No (Comparison group)	312	49.1	15	51.7	
Child's sex	Female	297	46.8	18	62.1	0.11
	Male	338	53.2	11	37.9	
Duration of pregnancy	Full term	475	76.0	13	72.2	0.70
	Less than full term	150	24.0	5	27.8	
Preparation of fruit & vegetables prior to consumption	Washed with treated water	117	18.4	6	20.7	0.12
	Washed with untreated water	402	63.3	22	75.9	
	Washed/disinfected with bleach or vinegar	116	18.3	1	3.4	
Total family income	From R\$ 0.00 to 100.00	48	7.7	4	14.3	0.40
	From R\$ 101.00 to 500.00	389	62.6	15	53.6	
	Above R\$ 500.00	184	29.6	9	32.1	

doi:10.1371/journal.pntd.0002943.t002

	Stage 1	_			Stage 2				Stage 3	~		
	Cistern	Cistern group (N ^a =332)	Compari	Comparison group (N = 332) Cistern group (N = 332)	Cistern g	iroup (N=332)	Compari	Comparison group (N = 332) Cistern group (N = 337)	Cistern	group (N= 337)	Compar	Comparison group (N=327)
	٩	%	c	%	5	%	c	%	۶	%	<u>د</u>	%
Positive	14	4.8%	31	10.9%	22	9.2%	38	16.7%	25	10.5%	17	7.6%
Vegative	278	95.2%	253	89.1%	217	90.8%	189	83.3%	214	89.5%	208	92.4%
Total of exams	292	100%	284	100%	239	100%	227	100%	239	100%	225	100%
^a N = initial number of monitored children.	er of monitor	ed children.										

.

1

doi:10.1371/journal.pntd.0002943.t003

Rainwater Cisterns and Giardiasis in Children

and dams. In recent studies conducted in other Brazilian regions with people from different age groups [1,7–10], the prevalence of G. duodenalis infections ranged from 10.6% to 32.1%.

The values reported in some of the studies cited above [1,8,10] may be underestimated because only one fecal sample was collected per child. It is known that the release of G. duodenalis cysts by the host occurs intermittently; therefore, at least three samples should be collected on alternate days, as was performed for the children analyzed in this study, to increase the detection sensitivity [38,39]. Nevertheless, Flanagan [38] warns about the possibility of a false-negative diagnosis.

In her review article, Flanagan [39] also notes that the two main risk factors associated with G. duodenalis infections are the consumption of unfiltered water that originated from surface water sources and, for children, placement in daycare centers. Additionally, Teixeira et al. [2] reported that the consumption of water from springs and wells significantly increased the odds of G. duodenalis infections in children who were one to five years old and living in informal settlement areas in Brazil.

As reported in the literature, the correlation between water consumption and pathogenic protozoa, especially G. duodenalis, infections has been demonstrated [2,40]. However, the question of what sources of water supply increase the risks of infection by these protozoa is still controversial because they have been detected in surface and underground sources and in rainwater. A multivariate analysis of the present study clarifies which factors affect the odds ratio of an infection by flagellate protozoans and in which direction.

After adjusting for socioeconomic, behavioral and hygiene habit variables, the odds ratio of a G. duodenalis infection were found to be 1.72 times higher in the children with no access to cisterns than among those from households with cisterns. This result is supported by the sedimentation effect of G. duodenalis cysts that may occur within cisterns, since water remains stored over a long period. Furthermore, a regular cleaning of the tanks, if performed as recommended, may further decrease the odds of ingesting cysts. As reported in the literature, many researchers have investigated the presence of cysts from this protozoan in rainwater harvesting cisterns but have not detected them [15,29,30].

In this study, we have used an intention-to-treat analysis. In the event, 16 families with cisterns (5%) at Stage 2 and 38 (11%) cistern families at Stage 3 were found during the field visits to have no water in their cisterns. It is possible that the association might have been stronger if we had taken account of cistern functionality.

The results of the present study have shown a possible protective effect of rainwater harvesting systems when compared to more insecure sources of water supply. However, other studies have shown an increase in the risk of G. duodenalis infection by rainwater ingestion, when comparing water from cisterns with sources likely to be safer. Hoque et al. [33] found a significant odds ratio of 8.3 for people that consumed roof harvested rainwater compared to those consuming water from the main water supply network of Auckland. Omar et al. [40] also found a significant odds ratio of 2.97 for giardiasis in people that consumed rainwater, rather than desalinated water.

Other variables were also independently correlated with G. duodenalis infections. An increased number of rooms per house was significantly associated with reduced odds of infection, which may be related to less interpersonal contact, thereby hindering the person-to-person transmission route, as already reported by many researchers regarding G. duodenalis infections [33,41,42].

Conversely, the number of rooms per house can also be linked to the economic situation of the family; thus, the result remains consistent when considering that wealthier families have better

Table 4. *G. duodenalis* infection in children in Northeast region of Minas Gerais State, Brazil, 2009–2010; **categorical** variables selected in the univariate analysis^a.

		OR ^b	95% CI	p-value
	Water supply			
ype of water supply	With a cistern	ref		
	Without a cistern	1.54	1.06-2.22	0.022
	Mother's health			
Pregnancy order	First	ref		
	Second or superior	1.46	1.01–.11	0.044
Duration of pregnancy	9 full months	ref		
	Less than 9 months	1.42	1.01-2.01	0.048
	Family structure			
Person responsible for child's care during the last year	Mother	ref		
	Another person	0.48	0.24-0.99	0.046
hild's caretaker's schooling	Illiterate	ref		
	Reads and writes	1.79	0.87-3.68	0.114
	Hygiene habits			
Hand hygiene prior to food preparation	Always	ref		
	With small frequency or never	3.50	1.63–7.53	0.001
Food hygiene prior to consumption	Washed with treated water	ref		
	Washed with untreated water	0.73	0.46-1.15	0.177
	Washed/disinfected with bleach or vinegar	1.01	0.60-1.69	0.977
	Socioeconomic characteristics			
otal family income	From R\$ 0.00 to 100.00	ref		
	From R\$ 101.00 to 500.00	0.59	0.33-1.04	0.068
	Above R\$ 500.00	0.59	0.32-1.08	0.088
	Sanitation conditions			
liver or stream close to the household	No	ref		
	Yes and the children have contact with the water	1.55	1.03–2.33	0.038
	Yes, but the children don't have contact with the water	1.04	0.7–1.55	0.834
	Solid waste and vectors			
Destination of household garbage	Burnt, buried or collected	ref		
	Open-air or thrown in the river	1.39	0.87-2.21	0.166
Flies/mosquitoes observed in the household throughout the year	Yes	ref		
	No	0.61	0.3-1.25	0.175

^athe univariate analysis associated each independent variable with the outcome and was adjusted only by the stage (time of data collection). ^bOR = Odds Ratio.

doi:10.1371/journal.pntd.0002943.t004

sanitation and hygiene conditions, which decreases the odds of infection with *G. duodenalis* cysts. This hypothesis was also confirmed by a household income analysis, which found lower odds of infection in children from families earning over R\$ 100.00 (about US\$ 40.00) a month. Similar results were reported by Machado et al. [43], who found (in a bivariate analysis) a correlation between a low socioeconomic status, which was represented by the family income and parental educational level, and *G. duodenalis* infections.

Regarding the "birth order", the firstborn children were confirmed to have lower odds of infection than those from subsequent births. This result may be correlated, once again, with the transmission via interpersonal contact but also with the greater care provided by mothers to their first child, minimizing the risk of infection for those children.

As expected, some hygiene habits were independently correlated with the outcome under analysis, such as improper hand hygiene prior to food preparation, which increased the odds of *G. duodenalis* infections by 4.78 times (Table 6). Proper hygienic habits are protective factors because they interrupt the fecal-oral transmission route of Giardia, as shown by Ratanapo et al. [44]. Regarding the variable "disinfection of fruit and vegetables," lower odds of infection were unexpectedly observed for individuals using untreated water, but this outcome may have been affected by the fact that 64% of the interviewees reported using untreated water. Only 18.6% and 17.6% of interviewees said they used **Table 5.** *G. duodenalis* infection in children in Northeast region of Minas Gerais State, Brazil, 2009–2010; continuous variables selected in the univariate analysis^a.

	OR ^b	95% CI	p-valor
Number of inhabitants per household	1.05	0.98–1.13	0.148
Number of rooms per household	0.90	0.81-1.01	0.068

^athe univariate analysis associated each independent variable with the outcome and was adjusted by the stage (time of data collection).

^bOR = Odds Ratio.

doi:10.1371/journal.pntd.0002943.t005

treated water and disinfected fruit and vegetables with bleach, respectively.

Finally, the "duration of pregnancy" was also selected in the multivariate model, which indicated a higher risk for preterm children and an association with socio-economic status. This result can be explained by the association of both giardiasis and preterm birth with low socio-economic status in Brazil [45,46] as well as an aggravated risk of infections in general for preterm infants.

Despite the higher odds of *G. duodenalis* infections for children with no access to water stored in cisterns, analyses of the microbiological quality of the water showed that the water consumed by the participants of both groups had similar and compromised quality, given the high rates of detection of total coliforms and *Escherichia* coli [32]. Regardless of its quality, the volume of 16,000 liters of water, which was available in cisterns in close proximity to the house, presumably enables the maintenance of more appropriate hygienic habits than the habits of individuals who must travel longer routes to collect water.

Graeff et al. [47] found that the amount of available water can affect the frequency of hand washing. They state that a mother needs approximately 20 liters of water to wash her hands after using the toilet and changing diapers prior to food preparation to

Table 6. Multivariate model for the analysis of *G. duodenalis* infection in children living in Northeast region of Minas Gerais State, Brazil, 2009–2010, according to socioeconomic, sanitary and pregnancy aspects.

	ORª	95% CI	p-value
Type of supply			
With cistern	ref	-	-
Without cistern	1.72	1.14–2.59	0.010
Stage			
Stage 1	ref	-	-
Stage 2	1.76	1.13-2.73	0.012
Stage 3	1.21	0.74–1.96	0.448
Birth Order			
First	ref	-	-
Subsequent	1.72	1.17–2.51	0.005
Duration of pregnancy			
9 full months	ref	-	-
Less than 9 months	1.70	1.19–2.43	0.004
Person who cooks washes his or her hands before starting activities			
Always	ref	-	-
With small frequency or never	4.78	1.95–11.76	0.001
Preparation of fruit & vegetables prior to consumption			
Washed with treated water	ref	-	-
Washed with untreated water	0.54	0.34-0.88	0.013
Washed/disinfected with bleach or vinegar	0.82	0.48-1.39	0.464
Total family income			
From R\$ 0.00 to 100.00	ref	-	-
From R\$ 101.00 to 500.00	0.48	0.26-0.88	0.018
Above R\$ 500.00	0.61	0.32-1.13	0.117
No. rooms per house	0.89	0.80-0.99	0.034

^aOR = Odds Ratio.

doi:10.1371/journal.pntd.0002943.t006

feed herself and the child. Thus, the volume of available water may have a greater impact on health than the quality of available water.

A major limitation of our study is that the allocation of cisterns was not a random process. As stated above, a set of selection criteria were applied, with a view to obtaining the maximum impact from the investment. These included (in descending order of importance) the lack of reliable alternative water sources, low household income, female household head, and relatively large numbers of children, disabled and elderly household members [13]. However, they were compared with a comparison group selected using the same criteria (except for having access to a cistern) and shown (Table 1) to have almost identical characteristics. Indeed, the selection of the comparison group practically exhausted the universe of eligible children in the two municipalities concerned. An additional control against bias arising from the allocation of cisterns is provided by the multivariate regression modelling (Table 6), which the only significant risk factors are among those which one would expect as biologically plausible, such as deficient hand or food hygiene.

The limitations of this study also include the difficulty in linking disease occurrence with the internal conditions of cement cisterns, which may promote or hinder the survival of *G. duodenalis* cysts. The parasite's unique physiological characteristics provide it with higher or lower odds of survival in certain locations. An analysis of the occurrence of the cysts in the water consumed by participants from both groups could also assist in the discussion of the findings. However, it was not possible to carry out such analysis because the tests for detection of *G. duodenalis* cysts in water samples are very expensive and require large volumes of water and specific equipment and reagents, absent in the municipalities where the study was conducted.

A third limitation is that the comparison group of children was not randomly selected. However, there is not much room for bias

References

- Basso RMC, Silva-Ribeiro RT, Soligo DS, Ribacki SI, Callegari-Jacques SM, et al. (2008) Evolução da prevalência de parasitoses intestinais em escolares em Caxias do Sul, RS. Rev Soc Bras Med Trop 41: 263–268.
- Teixeira JC, Heller L, Barreto ML (2007) *Giardia duodenalis* infection: risk factors for children living in sub-standard settlements in Brazil. Cad Saude Publica 23: 1489–1493.
- Prado MS, Strina A, Barreto ML, Oliveira-Assis AM, Paz LM, et al (2003) Risk factors for infection with *Giardia duodenalis* in pre-school children in the city of Salvador, Brazil. Epidemiol Infect 131: 899–906.
- Savioli L, Smith H, Thompson A (2006) Giardia and Cryptosporidium join the 'Neglected Diseases Initiative'. Trends Parasitol 22: 203–208.
- Prado MS, Cairncross S, Strina A, Barreto ML, Oliveira-Assis AM, et al. (2005) Asymptomatic giardiasis and growth in young children; a longitudinal study in Salvador, Brazil. Parasitology 131(Pt 1):51–6.
- 6. BRASIL. Ministûrio da Saû¤de. Secretaria de VigilûÂncia em Saû¤de. Departamento de AnûÂlise de SituaûÎûÊo de Saû¤de (2010) Saúde Brasil 2009: uma análise da situação de saúde e da agenda nacional e internacional de prioridades em saúde. Brasília. 368 p.
- Ferreira HS, Assunção ML, Vasconcelos VS, Melo FP, Oliveira CG, et al. (2002) Saúde de populações marginalizadas: desnutrição, anemia e enteroparasitoses em crianças de uma favela do "Movimento dos Sem Teto", Maceió, Alagoas. Revista Brasileira de Saúde Materna e Infantil 2: 177– 185.
- Andrade EC, Leite ICG, Vieira MT, Abramo C, Tibiriçá SHC, et al. (2011) Prevalência de parasitoses intestinais em comunidade quilombola no município de Bias Fortes, estado de Minas Gerais, Brasil, 2008. Epidemiologia e Serviços de Saúde 20: 337–344.
- Souza EA, Silva-Nunes M, Malafronte RS, Muniz PT, Cardoso MA, et al. (2007) Prevalence and spatial distribution of intestinal parasitic infections in a rural Amazonian settlement, Acre State, Brazil. Cad Saude Publica 23: 427– 434.
- Silva RR, Silva CAM, Pereira CAJ, Nicolato RLC, Negrão-Corrêa D, et al. (2009) Association between nutritional status, environmental and socioeconomic factors and *Giardia lamblia* infections among children aged 6–71 months in Brazil. Trans R Soc Trop Med Hyg 103: 512– 519.

because in the event almost all eligible children were selected, because fewer than expected eligible children were available.

Despite the limitations, the results confirm that having a cistern to store rainwater is associated with a reduction of G. duodenalis infections in children 4 months to 5 years of age. However, other risk factors, both environmental and family-related, were independently associated with the risk of Giardia infection It follows that interventions are needed which address not only the poor water quality and access of low income rural households, but also their hygiene behaviour, particularly hand hygiene [48].

Supporting Information

Table S1 Initial model for the analysis of *G. duodenalis* infection in children living in Northeast region of Minas Gerais State, Brazil, 2009-2010, according to socioeconomic, sanitary and pregnancy aspects and using marginal logistic regression (GEE). (DOCX)

Checklist S1 STROBE checklist. (DOC)

Acknowledgments

We are grateful for the support received from the Brazilian Semiarid Alliance (ASA), the rural population and the municipal administrations of Berilo and Chapada do Norte, and the non-governmental organizations ARAI (Rural Association of Child Welfare) and ACHANTI (Association of Assistance for the Needs of Workers and Children of Chapada do Norte).

Author Contributions

Conceived and designed the experiments: LH MC. Performed the experiments: JEF JLP. Analyzed the data: JEF EAC AGFCdC NBdS. Contributed reagents/materials/analysis tools: LEM. Wrote the paper: JEF MC LH JLP SC.

- Heller L, Bastos RKX, Vieira MBCM, Bevilacqua PD, Brito LLA, et al. (2004) Oocistos de *Cryptosporidium* e cistos de *Giardia*: circulação no ambiente e riscos à saúde humana. Epidemiol Serv Saúde 13(2): 79–92.
- IBGE. Instituto Brasileiro de Geografia e Estatística (2010) Censo Demográfico 2010. Disponível em http://www.ibge.gov.br/home/estatistica/populacao/ censo2010/default.shtm. Accessed 10 May 2011.
- BRASIL. Ministério do Desenvolvimento Social e Combate à Fome. Segurança Alimentar e Nutricional. Cisternas. Disponível em http://www.mds.gov.br/ falemds/perguntas-frequentes/seguranca-alimentar-e-nutricional/cisternas/ gestor/cisterna. Acessed 13 May 2011.
- BRASIL. Ministério do Desenvolvimento Social e Combate à Fome (2012) Cisternas. Disponível em http://www.mds.gov.br/segurancaalimentar/ acessoaagua/cisternas. Acessed 2 April de 2012.
- Rodrigo S, Leder K, Sinclair M (2009) Quality of stored rainwater used for drinking in metropolitan South Australia. Research Report N° 84. Adelaide: Water Quality Research Australia. 54 p.
- Melidis P, Akratos CS, Tsihrintzis VA, Trikilidou E (2007) Characterization of Rain and Roof Drainage Water Quality in Xanthi, Greece. Environ Monit Assess 127: 15–27.
- Andrade Neto CO. (2004) Proteção sanitária das cisternas rurais. In: Proceedings of the XI Simpósio Luso-Brasileiro de Engenharia Sanitária e Ambiental; 28 March–02 April 2004; Natal, Rio Grande do Norte. SILUBESA 2004. Available: http://www.aguadechuva.com/download/ protecaosanitariadascisternasruraisonofreccero.doc. Accessed 12 May 2014.
- Tomaz P (2003) Aproveitamento de água de chuva para áreas urbanas e fins não-potáveis. São Paulo. 180 p.
- Evans CA, Coombes PJ, Dunstan RH (2006) Wind, rain and bacteria: The effect of weather on the microbial composition of roof-harvested rainwater. Water Res 40: 37–44.
- Yaziz MI, Gunting H, Sapari N, Ghazali AW (1989) Variations in rainwater quality from roof catchments. Water Res 23: 761–765.
- Adeniyi IF, Olabanji IO (2005) The physico-chemical and bacteriological quality of rainwater collected over different roofing materials in Ile-Ife, southwestern Nigeria. Chem Ecol 21: 149–166.
- Chang M, Mcbroom MW, Beasley RS (2004) Roofing as a source of nonpoint water pollution. J Environ Manage 73: 307–315.

- Plazinska A. (2001) Microbiological Quality of Rainwater in Selected Indigenous Communities in Central Australia [Abstract]. In: Proceedings of the 10th International Conference on Rainwater Catchment Systems; 10–14 September; Mannheim, Germany. IRCSA 2001. Available: http://www.eng.warwick.ac. uk/ircsa/abs/10th/3_01.html. Accessed 12 May 2014.
- 25. Abbott S, Caughley B, Douwes J. (2007) The microbiological quality of roofcollected rainwater of private dwellings in New Zealand. In: Proceedings of the 13th International Conference on Rainwater Catchment Systems. August 2007; Sydney, Australia. IRCSA 2007. Available: http://www.ctahr.hawaii.edu/ hawaiirain/Library/papers/Abbots,%20Leder,%20Heyworth%20papers/ Abbott1.pdf. Accessed 12 May 2014.
- Ahmed W, Huygens F, Goonetilleke A, Gardner T (2008) Real-Time PCR Detection of Pathogenic Microorganisms in Roof-Harvested Rainwater in Southeast Queensland, Australia. Appl Environ Microbiol 74: 5490–5496.
- Ahmed W, Goonetilleke A, Gardner T (2010) Implications of faecal indicator bacteria for the microbiological assessment of roof-harvested rainwater quality in southeast Queensland, Australia. Can J Microbiol 56: 471–479.
- Ahmed W, Vieritz A, Goonetilleke A, Gardner T (2010) Health Risk from Potable and Non-Potable Uses of Roof-Harvested Rainwater Using Quantitative Microbial Risk Assessment in Australia. Appl Environ Microbiol 76: 7382–7391.
- Simmons G, Hope V, Lewis G, Whitmore J, Gao W (2001) Contamination of potable roof-collected rainwater in Auckland, New Zealand. Water Res 35: 1518–1524.
- Albrechtsen HJ (2002) Microbiological investigations of rainwater and graywater collected for toilet flushing. Water Sci Technol 46: 311–316.
- Crabtree KD, Ruskin RH, Shaw SB, Rosei JB (1996) The detection of Cryptosporidium oocysts and Giardia cysts in cistern water in the U.S.Virgin Islands. Water Res 30: 208–216.
- 32. Xavier RP (2010) Influência de barreiras sanitárias na qualidade da água de chuva armazenada em cisternas no semiárido paraibano. [MSc. Dissertation]. Campina Grande: Universidade Federal de Campina Grande. 114 p.
- Hoque ME, Hope VT, Scragg R, Kjellstrom T (2003) Children at risk of giardiasis in Auckland: a case-control analysis. Epidemiol Infect 131: 655–662.
- 34. Fonseca JE (2012) Implantação de cisternas para armazenamento de água de chuva e seus impactos na saúde infantil: um estudo de coorte em Berilo e Chapada do Norte, Minas Gerais. [MSc. Dissertation]. Belo Horizonte: Universidade Federal de Minas Gerais. 264p.
- 35. Fonseca JE, Silva CV, Pena JL, Heller L. (2011) A presença de sistemas de captação de água de chuva em cisternas tem favorecido o consumo de uma água com melhor

qualidade bacteriológica pelas famílias de zonas rurais? In: Proceedings of the 26th Congresso Brasileiro de Engenharia Sanitária e Ambiental; 25–29 September 2011. Porto Alegre, Rio Grande do Sul, Brasil. ABES 2011;

- PNUD. Programa das Nações Unidas para o Desenvolvimento Humano (2003) Atlas do Desenvolvimento Humano no Brasil. Brasil.
- Carvalho GLX, Moreira LE, Pena JL, Marinho CC, Bahia MT, et al. (2012) A comparative study of the TF-Test, Kato-Katz, Hoffman-Pons-Janer, Willis and Baermann-Moraes coprologic methods for the detection of human parasitosis. Mem Inst Oswaldo Cruz 107: 80–4.
- 38. Wolfe MS (1992) Giardiasis. Clin Microbiol Rev 5: 93-100.
- Flanagan PA (1992) Giardia diagnosis, clinical course and epidemiology. A review. Epidemiol Infect 109: 1–22.
- Omar MS, Mahfouz AAR, Moneim MA (1995) The relationship of water sources and other determinants to prevalence of intestinal protozoal infections in a rural community of Saudi Arabia. J Community Health 20: 433–440.
- Birkhead G, Vogt RL (1989) Epidemiologic surveillance for endemic Giardia lamblia infection in Vermont. The roles of waterborne and person-to-person transmission. Am J Epidemiol 129: 762–768.
- Keystone JS, Krajden S, Warren MR (1978) Person-to-person transmission of Giardia lamblia in day-care nurseries. Can Med Assoc J 119: 241–248.
- 43. Machado RC, Macrari EL, Cristante SFV, Carareto CMA (1999) Giardiasis and helminthiasis in children of both public and private day-care centers and junior and high schools in the city of Mirassol, São Paulo State, Brazil. Rev Soc Bras Med Trop 32: 697–704.
- 44. Ratanapo S, Mungthin M, Soontrapa S, Faithed C, Siripattanapipong S, et al. (2008) Multiple modes of transmission of giardiasis in primary schoolchildren of a rural community, Thailand. Am J Trop Med Hyg 78: 611–615.
- Fan RG, Portuguez MW, Nunes ML (2013). Cognition, behavior and social competence of preterm low birth weight children at school age. *Clinics* (Sao Paulo). 68: 915–21.
- 46. Silva RR, da Silva CA, de Jesus Pereira CA, et al. (2009). Association between nutritional status, environmental and socio-economic factors and Giardia lamblia infections among children aged 6–71 months in Brazil. Trans R Soc Trop Med Hyg. 103: 512–9.
- Graeff JA, Elder JP, Booth EM (1993)Communication for health and behaviour change: a developing country perspective. San Francisco. 204 p.
 Cairncross S, Valdmanis V (2006) Water supply, sanitation and hygiene
- Cairncross S, Valdmanis V (2006) Water supply, sanitation and hygiene promotion. In: Jamison DT, Breman JG, Measham AR, et al. (eds.) Disease Control Priorities in Developing Countries (2nd edition). Washington DC: The World Bank, Chapter 41, pp. 771–792.